

Limnology and Oceanography Letters 2022 © 2022 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/10/21.0272

## **CURRENT EVIDENCE**

# Causes and consequences of recent degradation of the Magdalena River basin, Colombia

Jorge Salgado <sup>(1,2\*</sup> Jonathan B. Shurin <sup>(1)</sup>, <sup>3</sup> María I. Vélez <sup>(1)</sup>, <sup>4</sup> Andrés Link <sup>(1)</sup>, <sup>5,6</sup> Laura Lopera-Congote, <sup>5,7</sup> Catalina González-Arango, <sup>5</sup> Fernando Jaramillo <sup>(1)</sup>, <sup>8,9</sup> Imenne Åhlén <sup>(1)</sup>, <sup>8</sup> Gabriela de Luna<sup>6</sup>

<sup>1</sup>Facultad de Ingeniería, Universidad Católica de Colombia, Bogotá, Bogotá, Colombia; <sup>2</sup>Centre for Environmental Geochemistry, School of Geography, University of Nottingham, University Park, Nottingham, UK; <sup>3</sup>Section of Ecology, Behavior and Evolution, University of California San Diego, La Jolla, California; <sup>4</sup>Department of Geology, University of Regina, Regina, Saskatchewan, Canada; <sup>5</sup>Department of Biological Sciences, Universidad de los Andes, Bogotá, Colombia; <sup>6</sup>Fundación Proyecto Primates, Bogotá, Colombia; <sup>7</sup>Department of Earth and Environmental Systems, Indiana State University, Terre Haute, Indiana; <sup>8</sup>Department of Physical Geography and Bolin Centre for Climate Research, Stockholm University, Stockholm, Sweden; <sup>9</sup>Baltic Sea Centre, Stockholm University, Stockholm, Sweden

## Scientific Significance Statement

Large tropical rivers provide essential societal benefits, including food production, irrigation, hydropower, transportation, and trade routes. However, these human activities pose unprecedented challenges to the integrity of the river systems, a problem that, in most cases, occurs without fully understanding the natural dynamics of these valuable ecosystems. Here, we synthesize the status of recent and long-term (decades-centuries) anthropogenic impacts occurring in one of the world's hotspots of tropical fish diversity, the Magdalena River, Colombia. Unfortunately, the Magdalena River has attracted little research attention compared to similar tropical rivers. We seek to promote research on the compounding impacts of climate change, river impoundment, alien invasive species, catchment deforestation, and water pollution on the Magdalena and the ecosystem services it provides.

## **Abstract**

The Magdalena River in Colombia is one of the world's largest (discharge =  $7100 \text{ m}^3 \text{ s}^{-1}$ ) tropical rivers, hosting > 170 aquatic vertebrate species. However, concise synthesis of the current ecological and environmental status is lacking. By documenting the anthropogenic stressors impacting the river on time scales ranging from centuries to decades, we found that the river system is subject to the compounding impacts of climate change, river impoundment, invasive alien species (IAS), catchment deforestation, and water pollution. We show that the Magdalena is a woefully understudied ecosystem relative to its critical importance to Colombia's economy, culture, and biodiversity compared with other similarly sized tropical rivers. We emphasize the need for research on (1) IAS population and ecological dynamics, (2) river damming and its links with IAS and climate change, and (3) land-use changes as well as identifying sources of water pollution and strategies for mitigation.

\*Correspondence: jorge.salgadobonnet@nottingham.ac.uk

<sup>a</sup>Author Contribution Statement: JS conceived the idea. JSA and FJ conceptualized the river dam section. JS and JBS conceptualized the invasive species section. MIV, LLC, CG, AL, and IA conceptualized the hydro-climatic section. FJ conceptualized the river Delta section. GL and JS conceptualized the water pollution section. JBS wrote the first draft. All authors contributed to the writing and editing of the manuscript.

**Data Availability Statement**: The Paleo data from Barbacoas and San Juan Lakes are available in the repository Pangea: https://doi.pangaea.de/10. 1594/PANGAEA.933157.

#### Associate editor: James B. Heffernan

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Large tropical rivers (> 1500-km long) provide essential benefits to society in terms of food production, irrigation, hydropower generation, transportation, and trade routes, among others (Bianchi 2016). However, several anthropogenic activities resulting in large-scale damming, water pollution, invasive alien species (IAS) introduction, deforestation and erosion, and climate change pose unprecedented ecological and environmental challenges to large rivers (Best 2019). Here, we document the status of recent and long-term (decades-centuries) anthropogenic activities that impact the Magdalena River basin in Colombia (northern South America) (Fig. 1), the prospectus for its continued importance as an economic and cultural center, and a hotbed of aquatic biodiversity.

The Magdalena River is one of the world's largest rivers, with a length of 1612 km and a discharge volume of 7100 m<sup>3</sup> s<sup>-1</sup> (Restrepo and Kjerfve 2000) (Table 1). The river dissects Colombia from South to North running between the Central and Eastern Andes Mountain range accumulating one of the highest global sediment yields (184 Mt yr<sup>-1</sup>) and the

largest in South America (Restrepo 2008). Since pre-Hispanic times, the Magdalena River has served as the central freshwater and food resource for a large human population and remains the major fluvial trade route in the country (Rodríguez 2015; Archila 2021). The river has forged Colombia's cultural and economic growth for millennia (Davis 2020; Archila 2021). Today, the Magdalena River basin hosts more than 30 million inhabitants, around 70% of the country's population, and accounts for 80% of the Gross Domestic Product (Dane 2022). A rich ethnic and cultural diversity is also reflected by the more than 15 indigenous groups that have occupied and shaped the territories during pre-historic, colonial, and modern times (Archila 2021) and the bv > 250 designated indigenous reservations corresponding to 34% of the nation's protected areas (IGAC 2012).

The Magdalena River is a hotspot of ecological significance, hosting more than 160 recorded fish species, of which 67 are endemic (Galvis and Mojica 2007; Rodríguez-Olarte et al. 2011;



**Fig. 1.** (a) Map of the Magdalena River basin showing the main river (blue line) and exemplary sites addressed in the text; and (b) examples of stressors, native species, and paleolimnological methods: (i) water hyacinth (*E. crassipes*); (ii) African common hippopo (*H. amphibius*); (iii) River tributary with high sediment load; (iv) dried areas of a floodplain lake during El Niño (ENSO); (v) the Magdalena catfish (*P. magdaleniatum*); (vi) the Magdalena turtle (*P. lewyana*); (vii) wetland fringe in Barbacoas; and (viii) Lake sedimentary core for paleolimnological studies.

River	Location	Length (km)	Drainage area (km²)	Discharge (m <sup>3</sup> s <sup>-1</sup> )	Fish spp. richness	Reference
Magdalena	South America	1612	257,438	7100	167	Restrepo and Kjerfve (2000)
Amazon	South America	6387	6,915,000	219,000	3000	Froese and Pauly (2022)
Orinoco	South America	2101	41,000	30,000	600	Froese and Pauly (2022)
Paraná	South America	3998	3,100,000	25,700	444	Froese and Pauly (2022)
Mekong	SE Asia	4023	810,000	16,000	877	Ziv et al. (2012)
Congo	Africa	4371	4,014,500	41,800	1044	Froese and Pauly (2022)
Nile	Africa	6690	2,870,000	5100	159	Froese and Pauly (2022)

Table 1. Hydrological and biotic characteristics of the Magdalena River and other similar tropical large (> 1500 km) rivers.

Jiménez-Segura and Lasso 2020) (Table 1). Two of these endemic species, the Magdalena prochilodontid (*Prochilodus magdalenae*) and the Magdalena catfish (*Pseudoplatystoma magdaleniatum*), undergo reproductive migrations (Zapata and Usma 2013) and have high socio-economic value for local and national fisheries (Jiménez-Segura and Lasso 2020) (Fig. 1). Furthermore, the river is within the Tumbes–Chocó–Magdalena biodiversity hotspot (Myers et al. 2000), holding other iconic aquatic and semi-aquatic vertebrates such as the American manatee (*Trichechus manatus*), the river otter (*Lontra longicauda*), and the Magdalena turtle (*Podocnemis lewyana*) (Angel-Escobar et al. 2014).

However, changes in land use, pollution, and expected impacts from climate change raise concerns for the future of the river's unique biodiversity. Since the 1970s, human population growth, the rising demand for food and land, water and hydropower, and mining have generated increasing stresses on the main river and its tributaries (Etter et al. 2008; Restrepo et al. 2015; Angarita et al. 2018). The middle and lower basins of the river have been subjected to pervasive deforestation and water contamination from oil and mineral extraction, household and industrial effluents, extensive cattle ranching, and, more recently, an expanding palm oil agribusiness (Restrepo and Escobar 2018). More than 20 large dam projects (> 200 MW of hydroelectric capacity) have also been built along the main channel and its tributaries, and new dam projects (> 80) are being implemented as the national and international demand for power continues to grow (Angarita et al. 2018). Expanding populations of aquatic IAS such as the Amazonian water hyacinth (Eichhornia crassipes), the Asian Basa catfish (Pangasianodon hypophthalmus), and the first worldwide case of feral African common hippo (Hippopotamus amphibius) outside Africa, further contribute to the current ecological fragility of the Magdalena River. The human population living on, and directly or indirectly influencing the Magdalena River, will also likely continue increasing within the coming decades (Tessler et al. 2015). Thus, identifying the many environmental and ecological stressors impacting the Magdalena River and their interactions, and developing effective strategies for their mitigation, are of urgent concern in Colombia and globally.

We discuss the main anthropogenic stressors impacting the aquatic system on time scales ranging from centuries to decades using contemporary and paleolimnological (the study of lake sediments to reconstruct past environmental conditions) approaches. We propose a multidisciplinary research initiative focusing on (1) the understanding of the ecology and impacts of aquatic IAS; (2) promotion of forest restoration and mitigation of deforestation; (3) reversal and prevention of eutrophication and other forms of water pollution; (4) maintenance of river flow regimes; and (5) instituting consistent long-term monitoring to assess ecosystem health and detect and understand the causes of critical thresholds of change in condition.

## Anthropogenic threats

#### **River damming and regulation**

Large dam projects pose one of the main threats to aquatic biodiversity in tropical rivers (Winemiller et al. 2016). Despite the value of water storage and electricity generation for promoting economic development, tropical river damming has far-reaching effects on the ecological integrity and biodiversity of rivers through water quality degradation and the restriction of movement of river flow, nutrients, major river-fed geochemical elements (e.g., Mg, Fe), and organisms (Poff et al. 2007) (Fig. 2). The impoundment of tropical rivers also affects the local climate (Degu and Hossain 2012), stimulating the growth of surrounding trees (Sun et al. 2021) and dampening natural hydrological variability (Jaramillo and Destouni 2015; Chaudhari and Pokhrel 2022). River damming further releases greenhouse gasses from sediments (Maavara et al. 2017) and increases evaporation levels from the water body's surface (Levi et al. 2015). Reservoir lentic conditions, when coupled with increases in nutrients (eutrophication), could additionally promote diatom species homogenization (Wengrat et al. 2018; Zorzal-Almeida et al. 2021), harmful algae blooms (HABs) (Labaut et al. 2020), and the invasion of multiple alien species (Johnson et al. 2008).

Hydropower infrastructure accounts for almost half of Colombia's electric demand in the Magdalena River basin,



**Fig. 2.** Schematic diagram summarizing some of the major impacts of tropical river damming, including enhancement of primary productivity and harmful algae blooms (HABs), biogeochemical cycles alteration, species invasion, the restriction of movement of water, nutrients, and organisms, changes in local climate, deforestation, the dampening of a natural river, salinization, forest flooding, and the release of greenhouse gasses. CI = chloride; DO = dissolved oxygen; Fe = iron; K = potassium; N = nitrogen; Na = sodium; P = phosphorus.

with an expected three-fold increase in the coming decades (Angarita et al. 2018). Appraisals of the impacts of these dams on the aquatic communities show that cumulative hydrologic alterations, coupled with the loss of longitudinal connectivity, will destabilize ecosystems by altering species composition and distribution, affecting the natural bimodal migration patterns, and reducing the extent of spawning habitats (Jimenez-Segura et al. 2022). Most fish endemism occurs across the tributary river headwaters; the longitudinal connectivity loss will pose a greater extinction risk for endemic species (Carvajal-Quintero et al. 2017). Basin-scale analyses on the lower river floodplains (Mojana; Fig. 1) further show that new hydropower infrastructure could drastically enhance (>90%) habitat fragmentation between lowland floodplains and upstream fish spawning habitats (Angarita et al. 2018; Jimenez-Segura et al. 2022). It could also intercept > 80% sediment load through the new dams (Angarita et al. 2018). Dam release floods also increase egg mortality of Magdalena's turtle (P. lewyana; Gallego-Garcia and Castaño-Mora 2014), while the introduction of alien fish species in the reservoirs as food sources, further stresses native fish communities (Jiménez-Segura and Lasso 2020; Zapata-Londoño et al. 2020). Therefore, current and planned damming projects will likely increase the number of > 100 threatened vertebrate species in the region (Jimenéz-Segura and Lasso 2020; IUCN 2022).

Data from Prado Reservoir (Upper Magdalena basin; Fig. 1) show that river impoundment leads to eutrophication (Salazar et al. 2002; Guevara et al. 2009) and the suppression of minimum and maximum peaks of runoff through the regulation of the natural flows of tributaries (Zamora et al. 2020). Although beneficial for energy production, agriculture, and

water management, the suppression of natural hydrologic variability further deprives many aquatic species of the conditions necessary for their reproduction, habitat, and migration (Carvajal-Quintero et al. 2017). In addition, eutrophication promotes HABs (Salazar et al. 2002; Roldán 2003), zooplankton community shifts from *Daphnia* sp. to *Bosmania longirostris* (Guevara et al. 2009), and the homogenization of physical and chemical conditions (Guevara et al. 2009). Data from Sogamoso Reservoir (middle basin; Fig. 1) further indicate substantial greenhouse gas emissions via organic matter decomposition at both in-filling (178,254.53-ton CO<sub>2</sub>eq year<sup>-1</sup>) and post-filling (466,946.57-ton CO<sub>2</sub>eq year<sup>-1</sup>) stages (Rodríguez and Peñuela 2022). The degradation of flooded forest organic material stored in large tropical reservoirs could take up to four to five decades (Campo and Sancholuz 1998; Salgado et al. 2020).

At the confluence with the Caribbean Sea, flow obstructions by dikes and sluices gates for flood control and agriculture have diminished runoff, and the river flows into the Magdalena River Delta (Jaramillo et al. 2018a). The deltas encompass alluvial plains and wetlands, mangrove forests, and one of the largest (4280 km<sup>2</sup>) freshwater-brackish lagoons of South America, known as the Cienega Grande de Santa Marta (CGSM). Hydrological estimations from the period 2007-2011 indicate that freshwater from the main Magdalena River now reaches the CGSM only 30% of the days (Jaramillo et al. 2018a); a result that is surprisingly low, given that the CGSM is a significant part of the delta of the Magdalena River. Although smaller tributaries from La Sierra Nevada de Santa Marta (Fig. 1) help maintain freshwater inputs into the CGSM, the temporal disconnection from the Magdalena River results in hypersaline conditions in many areas of the wetland (Jaramillo et al. 2018a).

#### Aquatic alien species

There are currently many (> 50) alien species introduced in the Magdalena River, including fish (e.g., the common carp Cyprinus carpio, the black bass Micropterus salmoides, the Yellow Tilapia Coptodon rendalli; and the Basa catfish P. hypophthalmus; Jimenéz-Segura and Lasso 2020), macrophytes (e.g., water hyacinth E. crassipes; and Brazilian Elodea Egeria densa; Carrillo et al. 2006; Salgado et al. 2019), mollusks (e.g., the bivalve Corbicula fluminea; De la Hoz-Aristizabal 2008), and mammals (water buffalo Bubalus bubalis; and African hippo H. amphibius; Shurin et al. 2020; Restrepo-Calle and Cadena 2021). Many of these have the potential to become or are already considered IAS (Gutierrez et al 2012; Jiménez-Segura and Lasso 2020). Among these aquatic IAS, the water hyacinth (E. crassipes), the Asian Basa catfish (P. hypophthalmus), the water buffalo (B. bubalis), and the African hippo (H. amphibius) are the most significant examples (Fig. 3). Hippos, for instance, were introduced into the middle basin of the Magdalena River in 1981 by the drug lord Pablo Escobar (Fig. 1). Two captive animals became feral following Escobar's death in 1993, and the population has grown steadily, around 10% per year (Subalusky et al. 2021). This invasion is the first-ever recorded case of feral hippos outside Africa, and informal estimates peg the population at around 100 animals which have recently been observed > 300 km north of where the population originated. Habitat modeling projected an eventual steadystate population of about 1500 animals by 2050 distributed throughout the middle and lower basins of the Magdalena (Castelblanco-Martínez et al. 2021).

In their natural range, hippos function as ecosystem engineers, vectoring nutrients acquired while grazing on land that is recycled as wastes in aquatic ecosystems (Stears et al. 2018). Hippo-driven excessive fertilization can lead to eutrophication, associated mass mortality of fishes, HABs, and disease vectors (Dutton et al. 2018). Comparing the trophic status of small artificial lakes with and without resident Colombian hippo populations confirmed that hippos may recapitulate their role as water fertilizers in their non-native range (Shurin et al. 2020). However, the results found statistically detectable but relatively minor differences in trophic status between hippo and non-hippo lakes. These differences were within the range of natural spatial variability among lakes. The lack of dramatic differences may partly be explained by more significant annual precipitation (compared to East African Savannas) and a comparable pervasive fertilizing impact of cattle ranching that simultaneously contaminates the studied lakes (Shurin et al. 2020). A lack of historical environmental and ecological information about the lakes in pre-hippos' time may also complicate the interpretation of the results. More research on the ecology of hippos in the wild is hence required. Important research foci to consider the genetic variation among the establishing population, the effects of hippos within the broader human-impact context of the river's catchment area (including water pollution and climate), and the links with other, more numerous introduced species with fewer media and scientific appeals such as cattle (*Bos taurus*) and water buffalo (*B. bubalis*).

Three different breeds of water buffalo have been introduced since the 1960s across the middle and lower basins of the Magdalena for the palm oil industry, meat, and dairy products (ICA 2017). Like hippos and cattle, buffalos are amphibious ecosystem engineers (Mihailou and Massaro 2021). By digging and wallowing on wetland margins, they can destabilize soil, expose large areas to evaporation, and reduce water retention during the dry periods (Skeat et al. 1996). Buffalos also suppress woody riparian vegetation by grazing on seedlings and saplings and promoting nutrient load and bioturbation (Petty et al. 2007). Herds of water buffalos are now commonly seen roaming the middle and lower basins of the Magdalena River and its floodplain lakes (e.g., Barbacoas Lake; Fig. 1). However, more research on their population dynamics and ecosystem effects in their non-native range is necessary.

Water hyacinth is considered among the 100 most invasive aquatic species worldwide, with annual management economic costs in South and Central America estimated at around USD 179.9 million (Heringer et al. 2021). Widespread records of this species have been reported across the Magdalena main reservoirs (e.g., Hidroituango, Porce II and III, Betania, and Prado), natural upland lakes (e.g., Fúquene Lake), lowland floodplain lakes (e.g., Barbacoas, Mojana), and in the river's main channel (Fig. 1). Water hyacinth's spread and dominance have been widely associated with lentic conditions and nutrient increases (Villamagna and Murphy 2010). Dense floating mats contribute to water anoxia by shading and preventing photosynthesis, impacting submerged biota, and producing decomposed organic matter (Salgado et al. 2019). Paleolimnological assessments (using plant macrofossils, pollen, and trace elements) from two<sup>210</sup>Pb dated cores in Lake Fúquene (Fig. 1) similarly showed that steadier hydrological conditions and increased nutrient loading stimulated Eichhornia proliferation over the last two centuries to its current levels of covering > 60% of the lake's surface area (Salgado et al. 2019). In reservoirs, the high catchment nutrient loading during the infilling process and the novel artificial lentic conditions create optimal conditions for Eichhornia to bloom (Thomaz et al. 2015). Historical and paleolimnological accounts in Brokopondo Reservoir, Surinam (Van Donselaar 1968), and the Panama Canal (Salgado et al. 2020) indicate that Eichhornia may, dominate reservoirs faster (1-5 years) than natural lakes. Costly remediation programs are required to control this nuisance species (Hearne 1966). Unfortunately, despite the ongoing expansion of hydropower and agricultural development in the Magdalena River, little attention has been paid to the invasion, ecology, and spread dynamics.

African hippo	E	Eutrophication	Bioturbidity	Oxygen drawdown	GHG emissions	HABs	Fish kills	Disea vecto	ase ring	
	Impact	(—)	(—)	(—)	(-)	(–)	(—)	(	)	
1-11	Evidence for MRB	LE*	LE*	LE*	LE*	LE*	NFE	NF	E	
		*In captivity observations								
<u>Water buffalo</u>		Eutrophicatio	n Bioturbation	Soil destabilizatio	Water retentior	GHG n emissions	Ripar woody ve	rian getation		
<u>(B. bubalis)</u>	Impact	(-)	(—)	(—)	(-)	(—)	(—	·)		
	Evidence for MRB	NE	LE	LE	NFE	NFE	NFE			
Water hyacinth		Oxygen N drawdown ren	Nutrients nobilization Na	C ivegation hom	ommunity ogenization	Community shift	Extinction risk	Habitat quality	Disease vectoring	
<u>[2: 0/000/</u>	Impact	()	(—)	(—)	(-)	(—)	(—)	(+/)	(—)	
×.	Evidence for MRB	LE	LE	LE	LE	LE	LE	LE	LE	
<u>Basa fish</u> ( <u>P. hypophtalamus)</u>		Bioturbation	Community homogenizatior	Community n shift	Extinction risk	Habitat quallity	Disease vectoring			
	Impact	(-)	(-)	(-)	(-)	(-)	(-)			
	Evidenc for MRB	e NFE	NFE	NFE	NFE	NFE	NFE			

**Fig. 3.** Schematic diagram of exemplary physicochemical (red) and ecological (green) impacts of the African hippo (*H. amphibius*), the water buffalo (*B. bubalis*), the water hyacinth (*E. crassipes*), and the Basa fish (*P. hypophthalmus*) on tropical rivers. The degree of evidence found in the literature for the Magdalena River basin (MRB) is categorized into strong evidence (HE; > 10 supporting references), medium evidence (ME; 6–10 supporting references); insufficient evidence (LE; < 5 supporting references) and not found evidence (NFE). Negative impacts are denoted by (–) and positive impacts by (+). GHG = greenhouse gasses; HABs = Harmful algal blooms.

Another spreading alien species is the Basa catfish (P. hypophthalmus), native to the Mekong River Delta in Vietnam (Southeast Asia). This migratory fish has a broad diet, a tolerance to extreme environmental conditions, and high reproductive and growth rates (Lakra and Singh 2010). The Basa fish is now one of the fastest-growing aquaculture crops in the tropics, being sold across > 100 countries (Singh and Lakra 2012). As reported for other tropical river systems (Garcia et al. 2018), it is believed that the Basa catfish were illegally introduced into the Magdalena River through local aquaculture programs and ornamental aquariums (Valderrama et al. 2016). The first captures in the wild date from 2015 across the main channel, tributaries, and floodplain lakes, and now it is rapidly expanding throughout the lowlands of the Magdalena basin (Valderrama et al. 2016). Studies from other tropical rivers have reported that Basa fish can impact native fish communities, affect habitat quality through bioturbation, and facilitate the transmission of fish pathogens (Thuy et al. 2010). A contributing factor to the potential spread of Basa toward likely invasive levels in the Magdalena River is the Colombian Decree 1780 issued by the Colombian Government (MADR and AUNAP 2015). This decree allows species used for fish farming purposes, whether introduced legally or

illegally to the country, to be declared "naturalized." Thus, they can become protected from conservational management strategies (Valderrama et al. 2016). We, therefore, urge studies on the Basa catfish ecology and impacts in the Magdalena River, as are also efforts to prevent its further spread and remove its "naturalized" status.

## Ecohydrology of the Magdalena River basin

El Niño–Southern Oscillation (ENSO) influences the hydroclimatological conditions of the Magdalena River basin, with extended periods of low precipitation (droughts) during El Niño events and prolonged rains (floods) during La Niña events (Restrepo and Kjerfve 2000). These climatological conditions are essential in connecting the main river and adjacent wetland and lake systems (Blanco et al. 2006; Jimenez-Segura et al. 2022) (Fig. 4). A paleolimnological reconstruction (diatoms and trace elements) from <sup>210</sup>Pb dated cores covering the last centuries of Barbacoas and San Juana Lakes (Fig. 1) showed lake surface area to vary according to ENSO and degree of connectivity to the Magdalena (Lopera-Congote et al. 2021). For instance, during a sequence of intense El Niño events between 1987 and 1992, Barbacoas Lake, directly



Fig. 4. Schematic diagram of the long-term (decades-centuries) paleolimnological evidence found for hydrological, ecological, and geochemical changes in the floodplain lakes (a) Barbacoas and (b) Sanjuana in response to ENSO events.

connected to the Magdalena, responded acutely during these dry years and was severely desiccated (Fig. 4). In turn, the surface area of the San Juana Lake, which is connected to the main river course through a series of tributaries (Fig. 4), lost only 10% of its historical value (Lopera-Congote et al. 2021). Similarly, in the delta region of the CGSM, the ENSO-driven periods of flood and drought influence the freshwater input from the Magdalena River into the CGSM wetland complex resulting in hypersaline conditions during dry events (Jaramillo et al. 2018*b*). Together these results suggest that during El Niño years, floodplain lakes with a higher degree of connectivity to the Magdalena River could face major droughts and substantial reductions in river flow, leaving them hydrologically isolated (Jimenez-Segura et al. 2022).

The Magdalena River and its tributaries have also experienced increasing sediment loads (Restrepo 2008). Fueled by industrialization and socioeconomic policies adopted in the 1970s, the Andean regions of the Magdalena River basin have witnessed unparalleled deforestation (Etter et al. 2008). Lowland savannas and floodplains have suffered equally strong transformations due to irrigation, replacing native forests and grasses by introducing pastures and crops (Etter et al. 2008). The recent globalization of agriculture has also resulted in large extensions of industrial agriculture (e.g., palm oil) across the central and lower basins of the river. Anthropogenic land use has transformed > 40 Mha of the river basin, accounting for 70% of national deforestation (Restrepo and Escobar 2018). Almost 69% of Andean forests and 30% of lowland forests were cut down by 2000, whereas between 2005 and 2010, deforestation in the river basin accounted for 24% of the combined deforestation in Colombia (Restrepo et al. 2015). All these developments contribute to sediment loading to the Magdalena.

Standing forests regulate land's hydrological and biogeochemical processes, ultimately determining the amount of organic matter, nutrients, and sediments reaching the river channel (Ellison et al. 2011; Ward et al. 2017). Uncontrolled land-use experiments comparing natural forest cover against standard agricultural and productive land covers show that deforestation in the Magdalena River basin could significantly reduce infiltration, enhance runoff from soil compaction by livestock trampling, and promote excess nutrient load from crops (Suescún et al. 2017). Estimates of sediment released by deforestation in the Magdalena basin indicate that between 1972 and 2010, erosion increased by more than 30% (Restrepo and Escobar 2018), from which 9% is loaded into the Magdalena through its tributaries. Between 2000 and 2010, the annual sediment load increased by 33% concerning pre-2000 conditions (Restrepo et al. 2015).

Land-use change in the Magdalena basin may exceed the rate of climate-driven erosion by several orders of magnitude (Restrepo and Escobar 2018). However, climate change and higher precipitation rates with more frequent extreme events (i.e., drought and flooding) may further magnify anthropogenic land-use impacts as an additional regional source of soil erosion. For instance, limestone and sand quarrying combined with erosive rain events have led to unprecedented loads of sediment into main tropical American rivers and estuaries (Jaramillo et al. 2016). Furthermore, rainfall will intensify under all climate change scenarios (IPCC 2021) on the Magdalena Andean slopes, with levels expected to increase by 5-20% by the end of the century. Seasonality is also projected to be more pronounced as the difference between the rainy and the dry seasons become larger, directly affecting the productivity of agricultural lands and natural ecosystems. More frequent consecutive rainfall events (IPCC 2021) would also lead to an increased risk of landslides, flooding, and soil erosion, increasing sediment load. All these climate changes are, thus, expected to amplify the effects of human-driven land-use changes on the Magdalena. Nonetheless, to the best of our knowledge, there has not been a study on

the hydro-climatic conditions of the river basin due to a lack of consistent baseline long-term discharge data (Åhlén et al. 2021). As such, attributing changes in runoff from the basin due to climate and land-use intensification is challenging.

Paleolimnological records offer one view of past sedimentological conditions and regimes of climatic variability in the absence of monitoring data. The reconstructions based on <sup>210</sup>Pb dated cores from Barbacoas and San Juana Lakes indicate a two-fold increase in sediment accumulation since 2000 (Lopera-Congote et al. 2021) (Fig. 4). Sediment accumulation in Barbacoas Lake (2-3 cm yr<sup>-1</sup>) was almost 10-fold higher than those of San Juana Lake  $(0.2-0.4 \text{ cm yr}^{-1})$  and the greatest reported among 10 other studied lakes across Central and North America (Yang et al. unpubl.). The parallel increases in accumulation rates post-2000 in Barbacoas and San Juana Lakes concurred during a wetter decade period that coincided with an increase in catchment deforestation (Restrepo et al. 2015; Restrepo and Escobar 2018). Most floodplain lakes in the middle and lower parts of the Magdalena River are shallow (< 4-m depth). Increases in erosional sediment load coupled with more frequent droughts and floods would likely double the natural rate of the lake infilling process and disconnection from the main channel (Restrepo and Kjerfve 2000). Disconnection from the main channel and reduced lake water levels may also enhance primary productivity and spread the already-occurring water hyacinth and other wetland flora (Fig. 2; Salgado et al. 2019). In addition, increases in sediment load and water hyacinth densities could promote suspended solids and water turbidity, oxygen drawdown, and alter food web structure (Mercer et al. 2014; Salgado et al. 2019).

One of Colombia's most significant infrastructure projects is to restore the navigability of the Magdalena River for large cargo ships by 2023. To this aim, the width and depth of the middle and lower basins of the river must be increased to allow navigability during the low water season. Such an engineering endeavor requires intensive dredging, and 1.4 million square meters of sediment have already been dredged (Cormagdalena 2021). In the Panama Canal, our paleolimnological assessments showed a four-fold increase in sedimentation rates with the significant expansion works to accommodate increasing global trade (Salgado et al. 2020). Thus, the sedimentation dynamics and hydrological connectivity between reservoirs, the Magdalena, and its floodplain lakes and delta will inevitably be even more altered, compromising the stability of these fragile ecosystems.

## Impacts on the Magdalena River Delta

The delta region of CGSM is part of a more extensive estuary system characterized by a wide variety of mangrove and swamp areas designated as an internationally important wetland under the Ramsar Convention (Gardner and Davidson 2011) and protected under a National Park designation in the *Parque Isla de Salamanca*. A mixture of coastal roads construction, flow regulation for agribusiness, urbanization, and pollution have led to the salinization and eutrophication of the CGSM system, resulting in a loss of more than 300 km<sup>2</sup> of mangrove coverage during the last 30 years; one of the most significant ever-recorded mangrove ecosystem losses in tropical Americas (Botero and Salzwedel 1999; Blanco et al. 2006; Restrepo et al. 2007; Vilardy et al. 2012; Röderstein et al. 2014). The collapse of this coastal ecosystem also corresponded with anomalous drought conditions (Jaramillo et al. 2018a,b). Paleoenvironmental reconstructions and ecosystem services risk analysis based on paleo-data and socioecological history have indicated that variation in salinity has been associated with the most significant temporal changes in the CGSM's aquatic ecosystems, including mangroves (Vilardy et al. 2011; Velez et al. 2014; Gutierrez-Cala 2020). Risk analysis pointed to hydrological connectivity with tributaries and the Caribbean as a management priority to minimize the risk of losing the services offered by the CGSM that benefit more than 300,000 people (Vilardy et al. 2011; Velez et al. 2018).

A lack of systematic river flow monitoring data from the Magdalena River and associated tributaries at the time of the mass mangrove mortality in the CGSM has hindered so far, the assessment of the overall ecosystem resilience (Zipper et al. 2020), and the partitioning of the specific roles of climate and anthropogenic activities on ecosystem change (Jaramillo et al. 2018*a*). Nevertheless, remote sensing data suggest a lateral connectivity fragmentation between the Magdalena River and the CGSM system through flow regulation (Jaramillo et al. 2018*a*). This hydrological fragmentation may have been responsible to a large extent for the current increases in salinity and eutrophication attributed to the death of the mangroves (Jaramillo et al. 2018*a*). Furthermore, radar interferometry— to construct maps of water level change show that currently, the Magdalena River delta,

including the CGSM, does not exhibit the typical flow conditions of similar unregulated floodplains in the country, such as the Atrato River in the western part of Colombia (Palomino-Ángel et al. 2019).

A restoration program for the CGSM is underway, aiming to reduce the salinity of current hypersaline areas of the lagoon by removing dead mangrove coverage and opening some channels. As of 2015, mangrove coverage had rebounded to approximately 400 km<sup>2</sup>, but the percentage cover has remained stagnant since 2011. Despite these restoration efforts, the CGSM is still in a bad environmental and ecological state and hence, it is included within the Montreaux Record (Hamman et al. 2019). The Montreaux Record is a list of Ramsar wetlands with priority conservation needs due to ongoing changes in ecological character resulting from human interference activities (Hamman et al. 2019).

## Water pollution and urbanization

Water pollution is an emerging threat of significant concern in the Magdalena River (Fig. 5), about which little evidence is available but whose effects likely contribute to further degrading the river ecosystem. The pollution is a straightforward consequence of a direct discharge of domestic and industrial sewage and runoff of nutrients and chemicals from agriculture, mining, and oil activities in the Magdalena River and its tributaries (Tejeda-Benitez et al. 2016). In the upper basin of the Magdalena, the city of Neiva is responsible for the first significant river pollution discharge due to a lack of a wastewater management system and oil and agricultural activities (Tejeda-Benitez et al. 2016). In the middle basin, the Magdalena River receives one of the significant pollutant contributors, the Bogotá River, which runs through the capital city of Bogotá before reaching the Magdalena. This tributary is

Geochemical		Water clarity		Eutrophication	GHG emission	Oxy s draw	gen down cont	Metal tamination	
<u>:</u>	Impact	(-	-)	(—)	(—)	(+,	/-)	(-)	
	Evidence for MRB	 L	E	ME	LE	L	=	LE	
Ecological		HABs	Species invasions	Communi s homogeniza	ty Com ation ទ	nmunity shift	Extinction risk	Habitat reduction	Foodweb health
	Impact	(-)	(—)	(—)		(-)	(—)	(—)	(—)
	Evidence for MRB	LE	LE	LE		LE	LE	LE	LE

**Fig. 5.** Schematic diagrams summarizing exemplary geochemical (red) and ecological (green) effects of water pollution on tropical river systems. The degree of evidence found in the literature for the Magdalena River basin (MRB) is categorized into strong evidence (HE; > 10 supporting references), medium evidence (ME; 6-10 supporting references); insufficient evidence (LE; < 5 supporting references) and not found evidence (NFE). Negative impacts are denoted by (-) and positive impacts by (+). GHG = greenhouse gasses; HABs = Harmful algal blooms.

among the topmost polluted rivers worldwide (Venegas et al. 2015; Díaz-Casallas et al. 2019). Physiological analyses of the aquatic biota from the Bogotá River have shown high bioaccumulation levels above the health standards of lead (Pb), chromium (Cr), and cadmium (Cd) (Forero et al. 2009). Mining activities coupled with extensive cattle ranching and agrofarms further promote pollution down the mouth of the Bogotá River, posing a high ecological risk to the associated biota in the middle and lower basins of the Magdalena River (Tejeda-Benitez et al. 2016).

Toxic metals, such as mercury (Hg), Pb, and arsenic (As), related to gold mining have been also found in the Magdalena catchment area, and to be incorporated into the aquatic food web (Mancera-Rodríguez and Álvarez-León 2006; Marrugo-Negrete et al. 2008a,b; Zapata et al. 2014). Mercury, for instance, is the third most toxic element for human health, according to the USA Agency for Toxic Substances and Disease Registry (Budnik and Casteleyn 2019). Its prevalence in water may also result in being re-emitted into the atmosphere or captured in the lake and river sediments (Kocman et al. 2017). Currently, methyl-mercury (Hg form in water) values in the lower basin of the Magdalena River are three to four times higher than pre-industrial times (1850) natural background values (Marrugo-Negrete et al. 2008a,b; Olivero-Verbel et al. 2015). The sources of this Hg pollution in the river are still debatable. Still, the evidence so far points toward a combination of a regional atmospheric deposition from cities and industry and from point sources of gold mining that have increased in the last two decades (Yang et al. unpubl.; Olivero-Verbel et al. 2015). In Barbacoas Lake, the paleo-sedimentary records indicate relatively high but constant Hg concentrations over the last six decades, with no clear indication of a recent significant increase in Hg related to mining activities (Yang et al. unpubl.). This is not surprising given that Barbacoas Lake is located south of the current gold mining hotspots (Northeast of Antioquia and South of Bolivar states; Fig. 1). Furthermore, the observed marked increase in sedimentation since the early 2000s and El Niño conditions may have also diluted the Hg concentrations in the lake sediments (Yang et al. unpubl.).

The widespread, uncontrolled application of nitrogen (N) and phosphorous (P) fertilizers and pesticides in agriculture and the aerial fumigation of herbicides as part of illegal drug eradication policies pose additional threats to the health of the Magdalena Basin ecosystem. Nitrogen and phosphorus biogeochemical cycles are completely disrupted, and current levels in the Magdalena exceed pre-1950 levels by one order of magnitude (Restrepo 2008). This nutrient excess affects many biological processes in riverine, terrestrial, and marine ecosystems (e.g., Ramírez-Valencia et al. 2021). Catchment urbanization across the Magdalena River basin is also expanding rapidly. Urbanization may change river discharge, reduce infiltration, and influence the delivery of organic matter enriched in nutrients, stimulating the growth of heterotrophic microorganisms (Kaushal et al. 2014; Wang et al. 2017). Transforming the river basin and hydrological characteristics may lead to eutrophication, hypoxia, HABs, and excess partial pressure of aquatic  $CO_2$  (Wang et al. 2017; Zhang et al. 2021). Projected urban development in the Magdalena River basin will thus likely increase water demand intensifying stagnant waters, oxygen drawdown, and potentially associated HABs among urban tributaries, especially during dry periods (Johnson et al. 2022).

## Concluding remarks and future research

Relative to its critical importance to Colombia's economy, culture, and biodiversity, the Magdalena River basin is a woefully understudied ecosystem compared to other similarly sized rivers worldwide. The limited evidence suggests that it is experiencing the compounding impacts of local and global environmental changes that may threaten ecological integrity and economic vitality (Tessler et al. 2015; Best 2019). The main emerging threats underpinning the social-ecological balance of the river system include large-scale damming, the spread of aquatic IAS, and the interconnected effects between catchment deforestation, erosion, and shifts in the hydro climatological characteristics of the basin. To understand these changes, our combined contemporary and paleolimnological research on the Magdalena River and associated floodplain lakes yield a more robust picture of variability over recent centuries and the main factors that destabilize and jeopardize the health of the community in the Magdalena River and tributaries. This information deficit hinders the development of strategic activities from reducing stresses and restoring ecosystem integrity, biodiversity, and ecosystem services (Tickner et al. 2020). To address these deficiencies, we recommend addressing the following knowledge gaps.

- Long-term studies of aquatic IAS are imperative to understand their distribution patterns, population dynamics (i.e., introduction, lag-phases, and exponential phase), and spread pathways (Strayer et al. 2006). Furthermore, research and management strategies must also be driven by scientific expertise (Restrepo-Calle and Cadena 2021) and by assessing many more species' economic, genetic, and ecological impact rather than focusing on species with a charismatic appeal (Beever et al. 2019). Consideration should be given to building scientific expertise in relevant areas, e.g., taxonomy of IAS, modeling pathways of spread, and ecological requirements of potential novel IAS (Sternberg et al. 2018).
- There is a need for more studies of river flow regimes for the natural functioning of the river channel, backwaters, and associated floodplain lakes (Chaudhari and Pokhrel 2022). Understanding ecological implications and societal effects (Moran et al. 2018), especially for indigenous communities of altered flow regimes, is also imperative (Anderson et al. 2019).

- A deeper understanding of how reforestation might serve as a mitigation strategy with collateral benefits for terrestrial, riparian, and aquatic biodiversity, climate regulation, flood control, human health, and local economies (Ellison et al. 2011). Modeling data and paleolimnological reconstructions show that deforestation degrades water quality and promotes hydrologic alteration and isolation of floodplain lakes from the river channel.
- Assess river ecosystem integrity based on continuous longterm monitoring data along the channel, its floodplains, and within the tributaries of subbasins. This will help detect and understand changes in the functioning of the river as a tightly connected hydrologic and ecological system and will provide reference baselines for delineation of management and conservation targets and objectives for restoration action plans (Poff et al. 2007).
- Understanding the consequences of extreme events, such as fluvial and coastal flooding or severe droughts, which act over short- and long-term time scales (Tessler et al. 2015). Assessing the intensity and distribution of hazardous events and how they interact with the other anthropogenic stressors is essential to determining future risk and the social and ecological integrity of the river delta (Santos and Dekker 2020).

Despite the ongoing anthropogenic issues stated here, we believe the Magdalena River is still a resilient system. However, to successfully rescue this unique and precious river system, it is imperative to address research knowledge gaps and strengthen inter-institutional coordination (e.g., RAMSAR-Government) for the management and decision making.

## References

- Åhlén, I., and others. 2021. Hydro-climatic changes of wetlandscapes across the world. Sci. Rep. 11: 1–11. doi:10. 1038/s41598-021-81137-3
- Anderson, E., and others. 2019. Understanding rivers and their social relations: A critical step to advance environmental water management. WIREs Water **6**: e1381. doi:10. 1002/wat2.1381
- Angarita, H., A. J. Wickel, J. Sieber, J. Chavarro, J. A. Maldonado-Ocampo, J. Delgado, and D. Purkey. 2018. Basin-scale impacts of hydropower development on the Mompós Depression wetlands, Colombia. Hydrol. Earth Syst. Sci. 22: 2839–2865. doi:10.5194/hess-22-2839-2018
- Angel-Escobar, D. C., S. Rodríguez-Buriticá, and M. C. Buitrago-Grisales. 2014. Support for the declaration of a public protected area in Barbacoas floodplain lakes, Municipality of Yondó, Antioquia., 1st Edition. Fundación Biodiversa.
- Archila, S. 2021. Inhabiting the Magdalena River during thepre-Hispanic period. In: Ospina et al. [ed.] Magdalena River: possible territories. Banco de la República, p. 121–156. s.

- Beever, E. A., D. Simberloff, S. L. Crowley, R. Al-Chokhachy, H. A. Jackson, and S. L. Petersen. 2019. Social–ecological mismatches create conservation challenges in introduced species management. Front. Ecol. Environ. 17: 117–125. doi:10.1002/fee.2000
- Best, J. 2019. Anthropogenic stresses on the world's big rivers. Nat. Geosci. **12**: 7–21. doi:10.1038/s41561-018-0262-x
- Bianchi, T. S. 2016, Deltas and humans. Oxford Univ. Press.
- Blanco, J., E. Viloria, and J. Narvaez. 2006. ENSO and salinity changes in the Cienaga Grande de Santa Marta coastal lagoon system, Colombian Caribbean. Estuar. Coast. Shelf Sci. **66**: 157–167. doi:10.1016/j.ecss.2005.08.001
- Botero, L., and H. Salzwedel. 1999. Rehabilitation of the Cienaga Grande de Santa Marta, a mangrove-estuarine system in the Caribbean coast of Colombia. Ocean Coast. Manag. 42: 243–256. doi:10.1016/S0964-5691 (98)00056-8
- Budnik, L. T., and L. Casteleyn. 2019. Mercury pollution in modern times and its socio-medical consequences. Sci. Tot. Environ. 654: 720–734. doi:10.1016/j.scitotenv.2018. 10.408
- Campo, J., and L. Sancholuz. 1998. Biogeochemical impacts of submerging forests through large dams in the Rio Negro, Uruguay. J. Environ. Manag. **54**: 59–66. doi:10.1006/jema. 1998.0222
- Carrillo, Y., A. Guarín, and G. Guillot. 2006. Biomass distribution, growth and decay of *Egeria densa* in a tropical highmountain reservoir (NEUSA, Colombia). Aquat. Bot. **85**: 7– 15.https://doi.org/10.1016/j.aquabot.2006.01.006
- Carvajal-Quintero, J. D., S. R. Januchowski-Hartley, J. A. Maldonado-Ocampo, C. Jézéquel, J. Delgado, and P. A. Tedesco. 2017. Damming fragments species' ranges and heightens extinction risk. Conserv. Lett. **10**: 708–716. doi: 10.1111/conl.12336
- Castelblanco-Martínez, D. N., and others. 2021. A hippo in the room: Predicting the persistence and dispersion of an invasive mega-vertebrate in Colombia, South America. Biol. Conserv. **253**: 108923. doi:10.1016/j.biocon.2020. 108923
- Chaudhari, S., and Y. Pokhrel. 2022. Alteration of River Flow and Flood Dynamics by Existing and Planned Hydropower Dams in the Amazon River Basin. Water Resour. Res. **58**: e2021WR030555. doi:10.1029/2021WR030555
- Cormagdalena (2021). Cormagdalena. [Accessed 2021 November]. Available from http://dc02eja.cormagdalena. gov.co/index.php?idcategoria=4650
- Degu, A. M., and F. Hossain. 2012. Investigating the mesoscale impact of artificial reservoirs on frequency of rain during growing season. Water Resour. Res. 48. doi:10.1029/ 2011WR010966
- DANE Departamento Administrativo Nacional de Estadística. 2022. Public bulletin: Departmental accounts Gross Domestic Product by department 2021 preliminary. [accessed 2022 June]. Available from https://www.dane.

gov.co/files/investigaciones/pib/departamentales/B\_2015/ Bol\_PIB\_dptal\_2021preliminar.pdf

Davis, W. 2020, Magdalena: River of Dreams. Random House.

- De La Hoz Aristizábal, M. V. 2008. First record in Colombia for *Corbicula fluminea* (Mollusca: Bivalvia: Corbiculidae), an invasive species. Boletín de Investigaciones Marinas y Costeras-INVEMAR **37**: 197–202.
- Díaz-Casallas, D. M., M. F. Castro-Fernández, E. Bocos, C. E. Montenegro-Marin, and R. González Crespo. 2019. 2008– 2017 Bogota River water quality assessment based on the water quality index. Sustainability **11**: 1668. doi:10.3390/ su11061668
- Dutton, C. L., A. L. Subalusky, S. K. Hamilton, E. J. Rosi, and D. M. Post. 2018. Organic matter loading by hippopotami causes subsidy overload resulting in downstream hypoxia and fish kills. Nat. Commun. 9: 1–10. doi:10.1038/s41467-018-04391-6
- Ellison, D., N. M. Futter, and K. Bishop. 2011. On the forest cover–water yield debate: From demand-to supply-side thinking. Glob. Change Biol. 18: 806–820. https://doi.org/ 10.1111/j.1365-2486.2011.02589.x
- Etter, A., C. McAlpine, and H. Possingham. 2008. Historical patterns and drivers of landscape change in Colombia since 1500: A regionalized spatial approach. Ann. Am. Assoc. Geogr. **98**: 2–23.
- Forero, A. R., J. F. G. Mantilla, and R. S. Martínez. 2009. Accumulation of lead, chromium, and cadmium in muscle of capitan (*Eremophilus mutisii*), a catfish from the Bogota River basin. Arch. Environ. Contam. Toxicol. **57**: 359–365. doi:10.1007/s00244-008-9279-2
- Froese, R., & Pauly, D. (2022). FishBase. World wide web electron publication. [accessed 2022 June 2]. Available from http://www.fishbase.org.
- Gallego-García, N., and G. Forero-Medina. 2014. Conservation of the Magdalena River Turtle in the Sinú River, Colombia. Testudo **8**: 1–12.
- Galvis, G., and J. Mojica. 2007. The Magdalena River fresh water fishes and fisheries. Aquat. Ecosyst. Health Manag. **10**: 127–139. doi:10.1080/14634980701357640
- Garcia, D. A., and others. 2018. The same old mistakes in aquaculture: The newly-available striped catfish *Pangasianodon hypophthalmus* is on its way to putting Brazilian freshwater ecosystems at risk. Biodivers. Conserv. **27**: 3545–3558. doi:10.1007/s10531-018-1603-1
- Gardner, R. C., and N. C. Davidson. 2011. The Ramsar convention, p. 189–203. *In, Wetlands*. Springer.
- Guevara, G., P. Lozano, G. Reinoso, and F. Villa. 2009. Horizontal and seasonal patterns of tropical zooplankton from the eutrophic Prado Reservoir (Colombia). Limnologica **39**: 128–139. https://doi.org/10.1016/j.limno.2008.03.001
- Gutierrez Cala, L. 2020. Looking for the present in the past: Social-Ecological Memory and Palaeoecology to explore changes in Ciénaga Grande de Santa Marta-Colombia. M. Sc. thesis, Stokolm University.

- Gutiérrez, F. D. P., C. A. Lasso, M. P. Baptiste, P. Duarte-Sánchez, and A. M. Díaz. 2012, *Catalogue of exotic and transplanted aquatic biodiversity in Colombia: Mollusks, crustacean, fish, amphibians, reptiles and birds*. Instituto de Investigación de Recursos Biológicos Alexander von Humboldt.
- Hamman, E., and others. 2019. Governance tools for the conservation of wetlands: The role of the Montreux Record under the Ramsar Convention. Mar. Freshw. Res. **70**: 1493–1502. doi:10.1071/MF18483
- Hearne, J. S. 1966. The Panama Canal's aquatic weed problem. Proceedings of the 19th Proceedings of Weed Science Conference V. **16**, p. 443–449.
- Heringer, G., and others. 2021. The economic costs of biological invasions in Central and South America: A first regional assessment. NeoBiota 67: 401–426. doi:10.3897/neobiota. 67.59193
- ICA. Instituto Colombiano Agropecuario. 2017. Censo Pecuario Nacional. [Accessed November 2021]
- IGAC-Instituto Geográfico Agustín Codazzi. 2012. Geographical map of indigenous safeguards. [Accessed November 2021] Available from https://geoportal.igac.gov.co/sites/ geoportal.igac.gov.co/files/geoportal/mapa\_resguardos\_ indigenas\_v1\_2012.pdf
- IPCC. 2021. Climate Change 2021: The Physical Science Basis. *In* V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J. B. R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou [eds.], *Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*. Cambridge Univ. Press.
- IUCN 2022. The IUCN Red List of Threatened Species. Version 2021-3. Available from https://www.iucnredlist.org/search? query=Magdalena%20River&searchType=species
- Jaramillo, F., and G. Destouni. 2015. Local flow regulation and irrigation raise global human water consumption and footprint. Science **350**: 1248–1251. doi:10.1126/science.aad1010
- Jaramillo, F., and others. 2018*a*. Assessment of hydrologic connectivity in an ungauged wetland with InSAR observations. Environ. Res. Lett. **13**: 024003. doi:10.1088/1748-9326/aa9d23
- Jaramillo, F., and others. 2018*b*. Effects of hydroclimatic change and rehabilitation activities on salinity and mangroves in the Ciénaga Grande de Santa Marta, Colombia. Wetlands **38**: 755–767. doi:10.1007/s13157-018-1024-7
- Jaramillo, F., M. Baccard, P. Narinesingh, S. Gaskin, and V. Cooper. 2016. Assessing the role of a limestone quarry as sediment source in a developing tropical catchment. Land Degrad. Dev. **27**: 1064–1074. doi:10.1002/ldr.2347
- Jimenéz-Segura, L., and C. A. Lasso [eds.]. 2020. XIX. Fish of the Magdalena River Basin, Colombia: Diversity, conservation and sustainable use. *In, Serie Editorial Recursos Hidrobiológicos y*

*Pesqueros Continentales de Colombia*. Alexander von Humboldt Biological Resources Research Institute.

- Jimenez-Segura, L. F., and others. 2022. Drivers for the artisanal fisheries production in the Magdalena River. Front. Environ. Sci.
- Johnson, J., and others. 2022. Whole-system analysis reveals high greenhouse-gas emissions from citywide sanitation in Kampala. Uganda Commun. Earth Environ. **3**: 1–10. doi: 10.1038/s43247-022-00413-w
- Johnson, P. T., J. D. Olden, and M. J. Vander Zanden. 2008. Dam invaders: Impoundments facilitate biological invasions into freshwaters. Front. Ecol. Environ. 6: 357–363. doi:10.1890/070156
- Kaushal, S. S., and others. 2014. Longitudinal patterns in carbon and nitrogen fluxes and stream metabolism along an urban watershed continuum. Biogeochemistry **121**: 23–44. doi:10.1007/s10533-014-9979-9
- Kocman, D., and others. 2017. Toward an assessment of the global inventory of present-day mercury releases to freshwater environments. Int. J. Environ. Res. Public Health 14: 138. doi:10.3390/ijerph14020138
- Labaut, Y., P. A. Macchi, A. A. Comas, C. R. Betancourt, and M. Díaz-Asencio. 2020. A 50-year sediment record of algal assemblage changes in Hanabanilla Reservoir. Cuba. J. Paleolim. 63: 235–250. doi:10.1007/s10933-020-00113-5
- Lakra, W. S., and A. K. Singh. 2010. Risk analysis and sustainability of *Pangasianodon hypophthalmus* culture in India. Aquac. Asia **15**: 34–37.
- Levi, L., F. Jaramillo, R. Andričević, and G. Destouni. 2015. Hydroclimatic changes and drivers in the Sava River Catchment and comparison with Swedish catchments. Ambio 44: 624–634. doi:10.1007/s13280-015-0641-0
- Lopera-Congote, L., J. Salgado, M. Isabel Vélez, A. Link, and C. González-Arango. 2021. River connectivity and climate behind the long-term evolution of tropical American floodplain lakes. Ecol. Evol. **11**: 12970–12988. doi:10.1002/ece3. 7674
- Maavara, T., R. Lauerwald, P. Regnier, and P. Van Cappellen. 2017. Global perturbation of organic carbon cycling by river damming. Nat. Commun. **8**: 1–10. doi:10.1038/ ncomms15347
- MADR and AUNAP. 2015. Ministry of Agriculture and Rural Affairs, and National Authority of Aquaculture and isheries of Colombia, Resolution 2287, 29 Dec 2015.
- Mancera-Rodríguez, N. J., and R. Álvarez-León. 2006. Current state of knowledge of the concentration of mercury and other heavy metals in freshwater fish in Colombia. Acta Biol. Colomb. **11**: 3–23.
- Marrugo-Negrete, J., L. N. Benitez, and J. Olivero-Verbel. 2008*a*. Distribution of mercury in several environmental compartments in an aquatic ecosystem impacted by gold mining in northern Colombia. Arch. Environ. Contam. Toxicol. **55**: 305–316. doi:10.1007/s00244-007-9129-7

- Marrugo-Negrete, J., J. O. Verbel, E. L. Ceballos, and L. N. Benitez. 2008*b*. Total mercury and methylmercury concentrations in fish from the Mojana region of Colombia. Environ. Geochem. Health **30**: 21–30. doi:10.1007/s10653-007-9104-2
- Mercer, E. V., T. G. Mercer, and A. K. Sayok. 2014. Effects of forest conversions to oil palm plantations on freshwater macroinvertebrates: A case study from Sarawak. Malaysia. J. Land Use Sci. 9: 260–277. doi:10.1080/1747423X.2013. 786149
- Mihailou, H., and M. Massaro. 2021. An overview of the impacts of feral cattle, water buffalo and pigs on the savannas, wetlands and biota of northern Australia. Austral Ecol. **46**: 699–712. doi:10.1111/aec.13046
- Moran, E. F., and others. 2018. Sustainable hydropower in the 21st century. Proc. Natl. Acad. Sci. USA **115**: 11891–11898. doi:10.1073/pnas.1809426115
- Myers, N., R. A. Mittermeier, C. G. Mittermeier, G. A. Da Fonseca, and J. Kent. 2000. Biodiversity hotspots for conservation priorities. Nature **403**: 853–858. doi:10.1038/ 35002501
- Olivero-Verbel, J., K. Caballero-Gallardo, and A. Turizo-Tapia. 2015. Mercury in the gold mining district of San Martin de Loba, South of Bolivar (Colombia). Environ. Sci. Pollut. Res. **22**: 5895–5907. doi:10.1007/s11356-014-3724-8
- Palomino-Ángel, S., J. A. Anaya-Acevedo, M. Simard, T. H. Liao, and F. Jaramillo. 2019. Analysis of floodplain dynamics in the Atrato River Colombia using SAR interferometry. Water **11**: 875. doi:10.3390/w11050875
- Petty, A. M., P. A. Werner, C. E. R. Lehmann, J. E. Riley, D. S. Banfai, and L. P. Elliott. 2007. Savanna responses to feral buffalo in Kakadu National Park, Australia. Ecol. Monogr. 77: 441–463. doi:10.1890/06-1599.1
- Poff, N. L., J. D. Olden, D. M. Merritt, and D. M. Pepin. 2007. Homogenization of regional river dynamics by dams and global biodiversity implications. Proc. Natl. Acad. Sci. USA 104: 5732–5737. doi:10.1073/pnas.0609812104
- Ramírez-Valencia, V., and others. 2021. Distribution of organic-walled dinoflagellate cysts in surface sediments of the southern Caribbean and the eastern tropical Pacific and its environmental implications. Mar. Micropaleontol. **167**: 102000. doi:10.1016/j.marmicro.2021.102000
- Restrepo, J. D. 2008. Applicability of LOICZ catchment-coast continuum in a major Caribbean basin: The Magdalena River, Colombia. Estuar. Coast. Shelf Sci. 77: 214–229. doi: 10.1016/j.ecss.2007.09.014
- Restrepo, J. D., and H. A. Escobar. 2018. Sediment load trends in the Magdalena River basin (1980–2010): Anthropogenic and climate-induced causes. Geomorphology **302**: 76–91. doi:10.1016/j.geomorph.2016.12.013
- Restrepo, J. D., and B. Kjerfve. 2000. Magdalena river: Interannual variability (1975–1995) and revised water discharge and sediment load estimates. J. Hydrol. **235**: 137–149. doi: 10.1016/S0022-1694(00)00269-9

- Restrepo, J., J. Blanco, C. Villamil, E. Viloria, J. C. Narvaez, and M. Rueda. 2007. Mangrove cover, fisheries, and environmental perturbations in the Cienaga Grande de Santa Marta (CGSM), Colombian Caribbean. Bull. Mar. Sci. 80: 931–932.
- Restrepo, J. D., A. J. Kettner, and J. P. Syvitski. 2015. Recent deforestation causes rapid increase in river sediment load in the Colombian Andes. Anthropocene **10**: 13–28. doi:10. 1016/j.ancene.2015.09.001
- Restrepo-Calle, S. R., and C. D. Cadena. 2021. Science denialism limits management of invasive hippos in Colombia. Front. Ecol. Environ. **19**: 323–325. doi:10.1002/fee.2373
- Röderstein, M., L. Perdomo, C. Villamil, T. Hauffe, and M. L. Schnetter. 2014. Long-term vegetation changes in a tropical coastal lagoon system after interventions in the hydrological conditions. Aquat. Bot. **113**: 19–31. doi:10.1016/j. aquabot.2013.10.008
- Rodríguez, M. 2015. Where is the Magdalena River going? Social, environmental and economic risks of the navigability project. Ediciones Uniandes.
- Rodríguez, D. C., and G. A. Peñuela. 2022. Estimation of greenhouse gas emissions of a tropical reservoir in Colombia. J. Water Clim. Change 13: 872–888.https://doi. org/10.2166/wcc.2022.330
- Rodríguez-Olarte, D., J. Mojica, and D. Taphorn. 2011. Chapter 15: Northern South America. Magdalena and Maracaibo Basins, p. 243–257. *In J. Albert and R. Reis [eds.]*, *Historical biogeography of neotropical freshwater fishes*. University of California Press.
- Roldán, G. 2003. [object HTMLSpanElement] Ecological implications of reservoirs implementation in Colombia. Rev. Univ. Catól. Oriente 16: 71–83.
- Salazar, C., F. Villa, and G. Reinoso. 2002. Spatial and temporal distribution of chlorophytes in Prado Reservoir (Tolima). Rev. Asoc. Colomb. Cienc. Biol. **14**: 57–64.
- Salgado, J., and others. 2019. Long-term habitat degradation drives Neotropical macrophyte species loss while assisting the spread of invasive plant species. Front. Ecol. Evol. **7**: 140. doi:10.3389/fevo.2019.00140
- Salgado, J., and others. 2020. A century of limnological evolution and interactive threats in the Panama Canal: Longterm assessments from a shallow basin. Sci. Total Environ. 729: 138444. doi:10.1016/j.scitotenv.2020.138444
- Santos, M. J., and S. C. Dekker. 2020. Locked-in and living delta pathways in the Anthropocene. Sci. Rep. **10**: 1–10. doi:10.1038/s41598-020-76304-x
- Shurin, J. B., and others. 2020. Ecosystem effects of the world's largest invasive animal. Ecology 101: e02991. doi: 10.1002/ecy.2991
- Singh, A. K., and W. S. Lakra. 2012. Culture of *Pangasianodon hypophthalmus* into India: Impacts and present scenario. Pak. J. Biol. Sci. **15**: 19–26.
- Skeat, A. J., J. East, and L. K. Corbett. 1996. Impact of feral water buffalo, p. 155–170. *In C. M. Finlayson and I. von*

Oertzen [eds.], Landscape and Vegetation Ecology of the Kakadu Region, Northern Australia. Kluwer Academic Publishers.

- Stears, K., and others. 2018. Effects of the hippopotamus on the chemistry and ecology of a changing watershed. Proc. Natl. Acad. Sci. USA **115**: E5028–E5037. doi:10.1073/pnas. 1800407115
- Sternberg, D., and B. Cockayne. 2018. The ongoing invasion of translocated sleepy cod (*Oxyeleotris lineolata*) in the Lake Eyre Basin, central Australia. Wildl. Res. **45**: 164–175. doi: 10.1071/WR17140
- Strayer, D. L., V. T. Eviner, J. M. Jeschke, and M. L. Pace. 2006. Understanding the long-term effects of species invasions. Trends Ecol. Evol. **21**: 645–651. doi:10.1016/j.tree.2006. 07.007
- Subalusky, A. L., and others. 2021. Potential ecological and socio-economic effects of a novel megaherbivore introduction: The hippopotamus in Colombia. Oryx 55: 105–113. doi:10.1017/S0030605318001588
- Suescún, D., J. C. Villegas, J. D. León, C. P. Flórez, V. García-Leoz, and G. A. Correa-Londono. 2017. Vegetation cover and rainfall seasonality impact nutrient loss via runoff and erosion in the Colombian Andes. Reg. Environ. Change 17: 827–839. doi:10.1007/s10113-016-1071-7
- Sun, L., and others. 2021. Exploring the influence of reservoir impoundment on surrounding tree growth. Adv. Water Resour. 153: 103946. doi:10.1016/j.advwatres. 2021.103946
- Tejeda-Benitez, L., R. Flegal, K. Odigie, and J. Olivero-Verbel. 2016. Pollution by metals and toxicity assessment using *Caenorhabditis elegans* in sediments from the Magdalena River. Colombia. Environ. Pollut. **212**: 238–250. doi:10. 1016/j.envpol.2016.01.057
- Tessler, Z. D., C. J. Vörösmarty, M. Grossberg, I. Gladkova, H. Aizenman, J. P. Syvitski, and E. Foufoula-Georgiou. 2015. Profiling risk and sustainability in coastal deltas of the world. Science **349**: 638–643. doi:10.1126/science. aab3574
- Thomaz, S. M., R. P. Mormul, and T. S. Michelan. 2015. Propagule pressure, invasibility of freshwater ecosystems by macrophytes and their ecological impacts: A review of tropical freshwater ecosystems. Hydrobiologia **746**: 39–59. doi:10. 1007/s10750-014-2044-9
- Thuy, D. T., P. Kania, and K. Buchmann. 2010. Infection status of zoonotic trematode metacercariae in Sutchicatfsh (*Pangasianodon hypophthalmus*) in Vietnam: Associations with season, management and host age. Aquaculture **302**: 19–25. doi:10.1016/j.aquaculture.2010.02.002
- Tickner, D., and others. 2020. Bending the curve of global freshwater biodiversity loss: An emergency recovery plan. Bioscience **70**: 330–342. doi:10.1093/biosci/biaa002
- Valderrama, M., J. I. M. Corzo, A. Villalba, and F. Ávila. 2016. Presence of basa fish, *Pangasianodon hypophthalmus*

(Sauvage, 1878) (Siluriformes: Pangasiidae), in the Magadalen River basin, Colombia. Biota Colomb. **17**: 98–104. doi:10.21068/C2016.v17n02a13

- Van Donselaar, J. 1968. Water and marsh plants in the artificial Brokopondo Lake (Surinam, S. America) during the first three years of its existence. Acta Bot. Neerl. **17**: 183–196.
- Velez, M., J. Escobar, M. Brenner, O. Rangel, A. Betancur, A. Jaramillo, J. Curtis, and J. Moreno. 2014. Middle to late Holocene relative sea level rise along the Colombian Caribbean coast inferred from a sediment core taken in the Ciénaga Grande de Santa Marta. The Holocene 24: 898–907. doi:10.1177/0959683614534740
- Velez, M. I., and others. 2018. Paleoenvironmental reconstructions improve ecosystem services risk assessment: Case studies from two coastal lagoons in South America. Water 10: 1350. doi:10.3390/w10101350
- Venegas, C., H. Diez, A. R. Blanch, J. Jofre, and C. Campos. 2015. Microbial source markers assessment in the Bogotá River basin (Colombia). J. Water Health 13: 801–810. doi: 10.2166/wh.2015.240
- Vilardy, S. P., J. A. González, B. Martín-Lopez, and C. Montes. 2011. Relationships between hydrological regime and ecosystem services supply in a Caribbean coastal wetland: A social-ecological approach. Hydrol. Sci. J. 56: 1423–1435. doi:10.1080/02626667.2011.631497
- Villamagna, A. M., and B. R. Murphy. 2010. Ecological and socio-economic impacts of invasive water hyacinth (*Eichhornia crassipes*): A review. Freshw. Biol. **55**: 282–298. doi:10.1111/j.1365-2427.2009.02294.x
- Wang, X., and others. 2017. pCO<sub>2</sub> and CO<sub>2</sub> fluxes of the metropolitan river network in relation to the urbanization of Chongqing. China. J. Geophys. Res. Biogeosci. **122**: 470–486. doi:10.1002/2016JG003494
- Ward, N. D., T. S. Bianchi, P. M. Medeiros, M. Seidel, J. E. Richey, R. G. Keil, and H. O. Sawakuchi. 2017. Where carbon goes when water flows: Carbon cycling across the aquatic continuum. Front. Mar. Sci. 4: 7. doi:10.3389/ fmars.2017.00007
- Wengrat, S., A. A. Padial, E. Jeppesen, T. A. Davidson, L. Fontana, S. Costa-Böddeker, and D. C. Bicudo. 2018. Paleolimnological records reveal biotic homogenization driven by eutrophication in tropical reservoirs. J. Paleolimn. 60: 299–309. doi:10.1007/s10933-017-9997-4
- Winemiller, K. O., and others. 2016. Balancing hydropower and biodiversity in the Amazon, Congo, and Mekong. Science **351**: 128–129. doi:10.1126/science.aac7082
- Zamora, D., E. Rodríguez, and F. Jaramillo. 2020. Hydroclimatic effects of a hydropower reservoir in a tropical

hydrological basin. Sustainability **12**: 6795. doi:10.3390/ su12176795

- Zapata, L. A., and J. S. Usma. 2013, *Guide to migratory species of Colombia's Biodiversity: Fish*, v. **2**. Ministerio de Ambiente y Desarrollo Sostenible y WWF-Colombia.
- Zapata, L. M., B. C. Bock, and J. A. Palacio. 2014. Mercury concentrations in tissues of Colombian Slider turtles, *Trachemys callirostris*, from northern Colombia. Bull. Environ. Contam. Toxicol. **92**: 562–566. doi:10.1007/s00128-014-1198-5
- Zapata-Londoño, M. N., E. J. Márquez, N. Restrepo-Escobar, and M. I. Ríos-Pulgarín. 2020. Population structure and reproduction of five fish species in a Neotropical reservoir. Revista Acad. Colomb. Ci. Exact. 44: 622–638. doi:10. 18257/raccefyn.1049
- Zhang, W., H. Li, Q. Xiao, and X. Li. 2021. Urban rivers are hotspots of riverine greenhouse gas (N<sub>2</sub>O, CH<sub>4</sub>, CO<sub>2</sub>) emissions in the mixed-landscape Chaohu lake basin. Water Res. **189**: 116624. doi:10.1016/j.watres.2020.116624
- Zipper, S. C., and others. 2020. Integrating the water planetary boundary with water management from local to global scales. Earths Future **8**: e2019EF001377. doi:10.1029/ 2019EF001377
- Ziv, G., E. Baran, N. So, I. Rodriguez-Iturbe, and S. A. Levin. 2012. Trading-off fish biodiversity, food security, and hydropower in the Mekong River Basin. Proc. Natl. Acad. Sci. USA 109: 5609–5614.https://doi.org/10.1073/pnas.1201423109
- Zorzal-Almeida, S., E. C. R. Bartozek, and D. C. Bicudo. 2021. Homogenization of diatom assemblages is driven by eutrophication in tropical reservoirs. Environ. Pollut. **288**: 117778. doi:10.1016/j.envpol.2021.117778

#### Acknowledgments

This study was funded by the Universidad Católica de Colombia through the "Convocatoria de Investigación 2021" awarded to Jorge Salgado (Project ID 493). The Baltic Sea Centre, Stockholm University, and the Swedish Research Council (VR) Project 2021-05774 have funded the contribution of Fernando Jaramillo. Programa de Investigación, Facultad de Ciencias (2022) has supported the contribution of Catalina González-Arango. María I. Vélez, and Laura Lopera-Congote were partially funded by an ELAP scholarship and NSERC DG.

## **Conflict of Interest**

The authors declare that they have no conflict of interest.

Submitted 09 February 2022 Revised 14 June 2022 Accepted 06 July 2022